Author’s response to reviews

Title: Incidence and characteristics of invasive fungal diseases in allogeneic hematopoietic stem cell transplant recipients: a retrospective cohort study

Authors:
Nicole Harrison (nicole.harrison@meduniwien.ac.at)
Margit Mitterbauer (margin.mitterbauer@meduniwien.ac.at)
Selma Tobudic (selma.tobudic@meduniwien.ac.at)
Peter Kalhs (peter.kalhs@meduniwien.ac.at)
Werner Rabitsch (werner.rabitsch@meduniwien.ac.at)
Hildegard Greinix (hildegar@greinix@klinikum-graz.at)
Heinz Burgmann (heinz.burgmann@meduniwien.ac.at)
Birgit Willinger (birgit.willinger@meduniwien.ac.at)
Elisabeth Presterl (elisabeth.presterl@meduniwien.ac.at)
Christina Forstner (christina.a.forstner@meduniwien.ac.at)

Version: 2 Date: 16 Dec 2015

Author’s response to reviews:
December 15th, 2015

Dear Editor,

We are submitting the second revision of our paper “Incidence and characteristics of invasive fungal diseases in allogeneic hematopoietic stem cell transplant recipients: a retrospective cohort study“ by Nicole Harrison, Margit Mitterbauer, Selma Tobudic, Peter Kalhs, Werner Rabitsch,
Hildegard Greinix, Heinz Burgmann, Birgit Willinger, Elisabeth Presterl, and Christina Forstner as original article to the Journal “BMC Infectious Diseases”.

We have carried out the following editorial revisions:

1. In the second revised version we have included a Conclusions section with a summary of our most important findings.

2. All authors have agreed to the addition of Prof. Birgit Willinger as a co-author and the signed forms have been sent to BMCseriesEditorial@biomedcentral.com.

We thank you for the review of our publication!

Sincerely,

Dr. Christina Forstner